Author's response to reviews

Title: Infective Endocarditis caused by staphylococcal endocarditis in a patient with Atopic dermatitis—a case report and literature review

Authors:

Gadha Mohiyiddeen (gaadha@yahoo.com)
Ian Brett (Ian.brett@tgh.nhs.uk)
Edward Jude (Edward.Jude@tgh.nhs.uk)

Version: 4 Date: 14 May 2007

Author's response to reviews:

Revisions are made based on the comments by the reviewers. Necessary changes have been made as suggested by the reviewers such as details of past history, more careful literature review on this topic.